Author's response to reviews

Title: Reading and writing difficulties in adolescence and later risk of welfare dependence. A ten year follow-up, the HUNT Study, Norway

Authors:

Kristine Pape (kristine.pape@ntnu.no)
Johan H Bjørngaard (johan.h.bjorngaard@ntnu.no)
Steinar Westin (steinar.westin@ntnu.no)
Turid L Holmen (turid.lingaas.holmen@ntnu.no)
Steinar Krokstad (steinar.krokstad@ntnu.no)

Version: 5 Date: 19 September 2011

Author's response to reviews:

New version containing the requested formatting changes only.